Author's response to reviews

Title: Haemoglobin, anaemia, dementia and cognitive decline in the elderly, a systematic review

Authors:

Ruth Peters (r.peters@imperial.ac.uk)
Lisa Burch (l.burch@imperial.ac.uk)
James Warner (j.warner@imperial.ac.uk)
Nigel Beckett (n.beckett@imperial.ac.uk)
Ruth Poulter (r.poulter@imperial.ac.uk)
Christopher Bulpitt (c.bulpitt@imperial.ac.uk)

Version: 7 Date: 5 August 2008

Author's response to reviews:

Punctuation amended slightly and the following words added:

In the first paragraph of Background, 'In the higher category this was statistically significant 17 percent increase' changed to 'In the higher category this was a statistically significant 17 percent increase'.

In the third paragraph of Background, 'Since that review the literature has grown and the very elderly become even more of an important group in society' changed to 'Since that review, the literature has grown and the very elderly have become even more of an important group in society'.